Author's response to reviews

Title: Infant skin-cleansing product versus water: A pilot randomized, assessor-blinded controlled trial

Authors:

Tina Lavender (tina.lavender@manchester.ac.uk)
Carol Bedwell (carol.bedwell@manchester.ac.uk)
Ediri O'Brien (Ediri.obrien@manchester.ac.uk)
Michael J Cork (M.J.Cork@sheffield.ac.uk)
Mark Turner (Mark.turner@liverpool.ac.uk)
Anna Hart (A.e.hart@lancaster.ac.uk)

Version: 7 Date: 10 May 2011

Author's response to reviews:

Please find uploaded a revised manuscript which contains a reformatted abstract. The manuscript has been proof read and minor edits made accordingly.

Kind Regards

Tina Lavender